





# 

**Citation:** Lo NC, Coulibaly JT, Bendavid E, N'Goran EK, Utzinger J, Keiser J, et al. (2016) Evaluation of a Urine Pooling Strategy for the Rapid and Cost-Efficient Prevalence Classification of Schistosomiasis. PLoS Negl Trop Dis 10(8): e0004894. doi:10.1371/ journal.pntd.0004894

**Editor:** Kelly K Baker, University of Iowa College of Public Health, UNITED STATES

Received: May 3, 2016

Accepted: July 11, 2016

Published: August 9, 2016

**Copyright:** © 2016 Lo et al. This is an open access article distributed under the terms of the <u>Creative</u> <u>Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** This study was funded by the American Society of Tropical Medicine and Hygiene Benjamin H. Kean Fellowship in Tropical Medicine (NCL), Doris Duke Charitable Foundation Clinical Research Mentorship grant (NCL, EB), European Research Council (JK), Infectious Diseases Society of America Medical Scholars program (NCL, JRA), Mount Sinai Hospital-University Health Network AMO Innovation Fund (IIB, JRA), and Stanford University Center for African Studies Graduate Student Fellowship (NCL). **RESEARCH ARTICLE** 

# Evaluation of a Urine Pooling Strategy for the Rapid and Cost-Efficient Prevalence Classification of Schistosomiasis

Nathan C. Lo<sup>1</sup>\*, Jean T. Coulibaly<sup>2,3,4,5</sup>, Eran Bendavid<sup>6,7</sup>, Eliézer K. N'Goran<sup>2,3</sup>, Jürg Utzinger<sup>4,5</sup>, Jennifer Keiser<sup>4,5</sup>, Isaac I. Bogoch<sup>8,9</sup>, Jason R. Andrews<sup>1</sup>

 Division of Infectious Diseases and Geographic Medicine, Stanford University School of Medicine, Stanford, California, United States of America, 2 Unité de Formation et de Recherche Biosciences, Université Félix Houphouët-Boigny, Abidjan, Côte d'Ivoire, 3 Centre Suisse de Recherches Scientifiques en Côte d'Ivoire, Abidjan, Côte d'Ivoire, 4 Swiss Tropical and Public Health Institute, Basel, Switzerland, 5 University of Basel, Basel, Switzerland, 6 Division of General Medical Disciplines, Stanford University, Stanford, California, United States of America, 7 Center for Health Policy and the Center for Primary Care and Outcomes Research, Stanford University, Stanford, California, United States of America, 8 Department of Medicine, University of Toronto, Toronto, Canada, 9 Divisions of Internal Medicine and Infectious Diseases, Toronto General Hospital, University Health Network, Toronto, Canada

\* nathan.lo@stanford.edu

# Abstract

## Background

A key epidemiologic feature of schistosomiasis is its focal distribution, which has important implications for the spatial targeting of preventive chemotherapy programs. We evaluated the diagnostic accuracy of a urine pooling strategy using a point-of-care circulating cathodic antigen (POC-CCA) cassette test for detection of *Schistosoma mansoni*, and employed simulation modeling to test the classification accuracy and efficiency of this strategy in determining where preventive chemotherapy is needed in low-endemicity settings.

# Methodology

We performed a cross-sectional study involving 114 children aged 6–15 years in six neighborhoods in Azaguié Ahoua, south Côte d'Ivoire to characterize the sensitivity and specificity of the POC-CCA cassette test with urine samples that were tested individually and in pools of 4, 8, and 12. We used a Bayesian latent class model to estimate test characteristics for individual POC-CCA and quadruplicate Kato-Katz thick smears on stool samples. We then developed a microsimulation model and used lot quality assurance sampling to test the performance, number of tests, and total cost per school for each pooled testing strategy to predict the binary need for school-based preventive chemotherapy using a 10% prevalence threshold for treatment.

### **Principal Findings**

The sensitivity of the urine pooling strategy for *S. mansoni* diagnosis using pool sizes of 4, 8, and 12 was 85.9%, 79.5%, and 65.4%, respectively, when POC-CCA trace results were

The funding organizations had no role in the design and conduct of the study; collection, management, analysis, and interpretation of the data; and preparation, review, or approval of the manuscript; or the decision to submit the manuscript for publication.

**Competing Interests:** The authors have declared that no competing interests exist.

considered positive, and 61.5%, 47.4%, and 30.8% when POC-CCA trace results were considered negative. The modeled specificity ranged from 94.0–97.7% for the urine pooling strategies (when POC-CCA trace results were considered negative). The urine pooling strategy, regardless of the pool size, gave comparable and often superior classification performance to stool microscopy for the same number of tests. The urine pooling strategy with a pool size of 4 reduced the number of tests and total cost compared to classical stool microscopy.

#### **Conclusions/Significance**

This study introduces a method for rapid and efficient *S. mansoni* prevalence estimation through examining pooled urine samples with POC-CCA as an alternative to widely used stool microscopy.

#### Author Summary

Schistosomiasis is a disease caused by parasitic worms that affects over 250 million people. The global control strategy is regular deworming of school-aged children. Before deworming campaigns can be conducted, one must know where the disease is present. The current method requires collection of individual stool and urine samples that are examined under a microscope by trained laboratory technicians. We present an alternative method that can inform where schistosomiasis is present in above 10% of the population, which is the threshold at which school-based deworming is recommended. The proposed strategy involves pooling multiple urine samples and using a rapid diagnostic test. The goal is to reduce the number of tests, cost, time, and laboratory infrastructure to guide decisionmaking. We collected data in Côte d'Ivoire to evaluate this new diagnostic procedure of pooling urine, and used computer simulation to predict its performance in classification of communities above or below the 10% threshold. We found that the urine pooling strategy with a pool size of 4 reduced the number of tests and cost compared to the current standard method, while maintaining the same accuracy. Our findings suggest that this strategy may be an effective and cost-saving method compared to traditional microscopy.

### Introduction

Schistosomiasis is a disease caused by parasitic worms of the genus *Schistosoma*, and affects over 250 million people residing in the world's poorest regions [1]. Historically, treatment programs have focused on control of disease morbidity [2–4]. However, the focus of these programs has recently shifted toward a goal of interrupting disease transmission and local elimination of helminth infections [4]. The World Health Organization (WHO) recommends a strategy of preventive chemotherapy (often known as 'mass drug administration') for control and elimination of schistosomiasis [2, 3]. The preventive chemotherapy program provides widespread empiric treatment with praziquantel, and traditionally focuses upon school-aged children [2, 3]. The frequency of preventive chemotherapy, if necessary, is based upon the infection prevalence [2, 3].

Most schistosomiasis control programs utilize prevalence estimates from survey samples of selected schools to guide treatment decisions for preventive chemotherapy [2, 3]. The WHO

currently recommends surveying one sentinel site (i.e., at least 50 children in one school) per 200,000–300,000 children in a homogeneous "ecological zone" [3]. For each child in a sentinel site, a fecal and urine sample is obtained. The stool is prepared using the Kato-Katz thick smear method and examined under a microscope (referred to as traditional stool microscopy) for the eggs of intestinal schistosomiasis (caused by *Schistosoma mansoni* and *S. japonicum*). The urine is tested for microhematuria or filtered and examined under a light microscope for the eggs of *S. haematobium*. These strategies are the current standard for prevalence estimation, but require laboratory infrastructure, trained personnel for slide preparation and interpretation, lack adequate sensitivity to detect low intensity infections, and are time- and resource-intensive [3, 5, 6].

These barriers have contributed to the challenge of mapping the prevalence of schistosomiasis in endemic countries, which is necessary to inform where preventive chemotherapy should be implemented. Furthermore, a high-resolution understanding of local prevalence is especially important for schistosomiasis compared to other helminth infections, because schistosomiasis is highly geographically focal due to dependence on freshwater for completion of the life cycle [7].

The pooling of urine samples with the point-of-care circulating cathodic antigen (POC-CCA) cassette test may provide a cost-efficient alternative to traditional stool microscopy. The pooling of biological specimens (including feces and urine) as an efficient methodology for disease screening is common across veterinary parasitology, HIV, and other diseases [8, 9]. The POC-CCA test is a rapid diagnostic test that uses urine for the binary detection of S. mansoni. Most notably, the POC-CCA test retains high sensitivity at low intensity infections and does not read positive once an infection is resolved making it suitable for application in pooled sampling [5, 10-12]. However, there are currently no studies that have examined the diagnostic characteristics, optimal pooling size, or cost-efficiencies associated with a urine pooling methodology using the POC-CCA cassette test. Furthermore, since the preventive chemotherapy strategy is based upon a broad prevalence categorization (WHO groups settings by <10%, 10– 50%, or >50% prevalence) rather than a specific prevalence value, a simplified classification tool can be employed. Lot quality assurance sampling (LQAS) is an approach to evaluate the accuracy of classification of unknown entities into binary or multiple groups, according to predefined thresholds [13–15]. This tool has been well characterized in the helminthiasis literature [13–15], and provides an attractive option for estimating the broad prevalence category to reduce number of tests and cost.

To address this critical need, we conducted an empirical evaluation of the accuracy of urine pooling with the POC-CCA cassette test, and then applied simulation modeling using a LQAS framework to evaluate classification accuracy and efficiency for informing targeted treatment of schistosomiasis.

#### Methods

#### **Ethics Statement**

This study was approved by the Institutional Review Board for Human Subjects Research at Stanford University School of Medicine (Stanford, CA, United States of America) and ethical clearance was obtained from the Ministry of Health and Public Hygiene of Côte d'Ivoire (CNER, reference no. 037/MSLS/CNER-dkn). We obtained written informed consent from parents/guardians and oral assent from children in Azaguié Ahoua, Côte d'Ivoire. All data were coded and treated as confidential personal health information. After completion of the study, all study participants received praziquantel (single 40 mg/kg oral dose) and albendazole (single 400 mg oral dose) at no cost as per national guidelines [2, 3].

#### Study Setting and Population

This cross-sectional study was performed in August 2015 at schools from six neighborhoods in the Azaguié health district (geographic coordinates: 5° 37' 40" N latitude and 4° 5' 12" W longitude) of Côte d'Ivoire, located at 40 km north from Abidjan. These settings ranged from moderate to high endemicity for *S. mansoni* to ensure a sufficient number of positive urine and stool samples were obtained. We selected 180 children between the ages of 6 and 15 years, informed by sample size estimations for diagnostic tests (<u>S1 Text</u>) [16].

#### Field and Laboratory Procedures

A detailed census was carried out in early August 2015 to determine the number of schoolaged children per neighborhood in Azaguié Ahoua village. Based on that list, 30 children were randomly selected per neighborhood. The purpose and procedures of the study were explained to the village and health authorities. For children who provided oral assent, and whose parent/ guardian provided written informed consent, we obtained two stool and two midstream urine samples over two days. Urine was collected between 10:00 and 12:00 hours.

For both stool sample collected over consecutive days, duplicate Kato-Katz thick smears were prepared the same day as collection following standard protocol, for a total of four Kato-Katz thick smears per child [17]. Slides were labeled with a de-identified code and read by experienced laboratory technicians using light microscopy. The presence and quantity of helminth eggs was counted on each slide for *S. mansoni*. For quality control purposes, we randomly selected 10% of the Kato-Katz thick smears, including both positive and negative slides, for re-examination by a senior technician. If discrepancies above the tolerance margin were noted, the results were discussed with the technicians and the slides were read a third time to reach agreement (S1 Text).

All of the first day urine samples were tested using the POC-CCA cassette test (Rapid Medical Diagnostics; Pretoria, South Africa, batch #50174) on the same day of sample collection. To perform the test, one drop of urine was placed into the POC-CCA cassette well followed by one drop of the test buffer. Two blinded study personnel experienced with POC-CCA read the tests independently 20 min after the addition of buffer. In cases of disagreement, a third blinded technician read the results and a decision was made based on agreement of at least two out of three individuals. Tests were read as negative, trace positive, 1+, 2+, or 3+ according to the color intensity of the test band, and tests that did not develop the control band were repeated.

For the pooling of urine samples, we first identified individuals as positive or negative using quadruplicate Kato-Katz thick smears and a single POC-CCA test. Positive samples were from children with a positive Kato-Katz and POC-CCA test, while negative samples were from children with a negative Kato-Katz and POC-CCA test. Hence, we only included samples with concordant results from both tests. One positive urine sample (~5 ml) was then combined with equal volumes of three (n = 4), seven (n = 8), or 11 (n = 12) negative urine samples. All samples were poured into a urine collection container, and a brief mixing step was done with the disposable pipette provided with the POC-CCA test kit. We then performed the POC-CCA test as described above.

#### **Statistical Analysis**

We calculated the sensitivity and specificity of quadruplicate Kato-Katz thick smears and the individual POC-CCA test using latent class analysis [18]. This analytical strategy combines prior knowledge on sensitivity and specificity and the observed data to simultaneously calculate point estimates and 95% credible intervals around the sensitivity and specificity for two or more diagnostics tests without assuming any test as the 'gold' standard [18]. We used the

Gibbs sampler model for two diagnostic tests with assumption of independence, and derived a prior distribution following consensus from literature using a beta distribution (S1 Text) [5, 10, 12, 18, 19]. Only children with complete data records (i.e., quadruplicate Kato-Katz thick smears and POC-CCA test) were included in the final analysis. We calculated eggs per gram of feces (EPG) from the four Kato-Katz thick smears using an arithmetic mean and the conventional 24-fold multiplier [20]. Sensitivity of the pooled samples was calculated assuming that a single positive urine sample classified the entire pool as positive. We used a logistic regression to model the sensitivity of pooled urine samples (with pool sizes of 4, 8, and 12) as a function of the EPG of the one infected urine. For each individual, we defined the independent variable as the arithmetic mean EPG from quadruplicate Kato-Katz thick smears and the dependent binary outcome as whether or not the pooled urine was read as positive with the POC-CCA cassette test. All analyses were conducted by considering the POC-CCA trace result as positive (POC-CCA(tr+)) and negative (POC-CCA(tr-)) to examine the impact of this interpretation on results [2, 3].

#### Model and Assumptions

We developed an individual-level stochastic decision analytic model (first order Monte Carlo simulation; microsimulation) for diagnosis of schistosomiasis to test the performance of the urine pooling strategies using the LQAS classification tool. We compared five strategies: (i) duplicate Kato-Katz thick smears (WHO standard); (ii) single POC-CCA test; (iii) pooled POC-CCA test (n = 4); (iv) pooled POC-CCA test (n = 8); and (v) pooled POC-CCA test (n = 12). A simulated cohort of 100,000 individuals was created for each prevalence value (0-20%), with assignment of infection status and EPG (when applicable) to each person. We assumed a negative binomial statistical distribution, and used a specified prevalence and inferred infection intensity to simulate the distribution of egg counts in the cohort (S1 Text) [21].

In the microsimulation, a sample of individuals was randomly selected from the simulated cohort. Each individual was assigned to one of four mutually exclusive states (true positive, true negative, false positive, or false negative) using the computed sensitivity and specificity of the respective diagnostic strategy and knowledge of an individual's "true" infection status (from simulated egg counts). The sensitivity (in relation to infection intensity) and specificity for duplicate Kato-Katz thick smears for individual samples was derived from literature [5], while we used the sensitivity (irrespective of infection intensity) and specificity from the latent class analysis for single POC-CCA. For pooled POC-CCA test strategies with more than one positive sample, we conservatively used the total EPG count to relate the infection intensity to our pooling data (in which only one urine was infected) and computed sensitivity using the logistic model (S1 Text; Fig A1-A2 in S1 Text). To address the impact of urine pooling on specificity, we used the POC-CCA specificity estimate from the Bayesian latent class analysis (which assumes no diagnostic 'gold' standard), and accounted for the increased probability of including a false positive by adding more urine samples that could be false positives. We assumed the effect of diluting a false positive urine sample was comparable to diluting a lightintensity infected urine sample (S1 Text). Ultimately, the dilution effect counteracted the potential for decreased specificity from sample pooling (<u>S1 Text</u>).

As preventive chemotherapy strategies are based on classifying regions into a prevalence bin (i.e., above or below a predefined prevalence threshold), we chose our primary outcome as the probability of correct binary classification around one prevalence threshold. Specifically, we tested the WHO-recommended prevalence threshold for preventive chemotherapy against schistosomiasis (10% prevalence) [2, 3]. We defined classification certainty as the proportion

of correctly categorized settings within 5% of the prevalence threshold, following estimates from prior studies [14]. The decision rule was chosen based on the median number of positive tests from the microsimulation at the prevalence (10%; <u>S1 Text</u>). The microsimulation was run 10,000 times for each strategy using a range of 15 to 500 tests for each strategy. We estimated the number of correct categorical classifications for each strategy at all prevalences. We also tested the potential to reduce the number of tests with the urine pooling strategies compared to stool microscopy, while maintaining the same level of classification accuracy. We compared our results against previous analytical approaches to LQAS [14], which did not account for imperfect test accuracy. A loess algorithm was applied for visualization and interpolation.

The cost of each strategy was estimated from recent literature, incorporating costs for supplies, labor, and pooling. We added an additional US\$ 0.50 per extra sample in the pooling strategy to account for personnel time and collection container costs (<u>S1 Text</u>) [6]. The total cost for POC-CCA pool sizes of 4, 8, and 12 urines was estimated at US\$ 6.63, US\$ 8.63, and US\$ 10.63, respectively; Kato-Katz was estimated at US\$ 3.99 (<u>S1 Text</u>).

Data were recorded in a Microsoft Excel spreadsheet, and statistical analysis and data visualization were performed with Python and R 3.1.1 (R Foundation for Statistical Computing; Vienna, Austria). The authors support the importance of data sharing and transparency in research; hence, full model code and data are available upon request to the corresponding author.

#### Sensitivity Analysis

We performed a series of sensitivity analyses to assess the robustness of our findings. We conducted one-way sensitivity analyses to examine the effect of individual parameters on the total cost of the urine pooling strategy for a 90% level of certainty in classification. We varied number of tests, cost estimates, sensitivity, specificity, and the dilution effect on light infections. We also tested the robustness of results against setting-specific epidemiologic differences, where the EPG distribution was varied for the same prevalence. We repeated the main analysis using a lower proposed prevalence threshold based on recent cost-effectiveness modeling (5% prevalence instead of the current 10% WHO cutoff) [21, 22].

#### Results

#### Sensitivity and Specificity of Individual and Pooled Diagnostic Strategies

From the 114 school-aged children with complete data, 59.6% were positive by quadruplicate Kato-Katz thick smears, 69.3% were positive by single POC-CCA(tr+), and 50.0% were positive by single POC-CCA(tr-) (Table 1). Using a latent class model, we estimated the sensitivity and specificity of quadruplicate Kato-Katz thick smears at 78.2% (95% CI: 71.0–84.5%) and 96.1% (95% CI: 90.8–98.8%), respectively (Table 2). We estimated the sensitivity of POC-CCA(tr+) test at 95.1% (95% CI: 89.2–98.5%) and POC-CCA(tr-) test at 74.4% (95% CI: 63.8–83.7%). The specificity of POC-CCA(tr+) test was 82.9% (95% CI: 71.5–91.6%) and POC-CCA(tr-) test was 94.5% (95% CI: 86.9–98.5%).

The overall sensitivity of the urine pooling strategy (POC-CCA(tr+)) was 85.9%, 79.5%, and 65.4% for pool sizes of 4, 8, and 12 (<u>Table 3</u>). For the urine pooling strategy (POC-CCA(tr-)), the sensitivity was 61.5%, 47.4%, and 30.8% for pool sizes of 4, 8, and 12. The modeled specificity ranged from 94.0–97.7% for the urine pooling strategies. The sensitivity of each strategy increased for detection of moderate and heavy infections.

The sensitivity of each pooled strategy (POC-CCA(tr-)) was modeled as a logistic function with respect to infection intensity, measured in EPG (Fig 1). Sensitivity was strongly positively

|                                  | Setting A | Setting B | Setting C | Setting D | Setting E | Setting F | Overall |
|----------------------------------|-----------|-----------|-----------|-----------|-----------|-----------|---------|
| Number of children               | 20        | 12        | 16        | 22        | 21        | 23        | 114     |
| Kato-Katz <sup>a</sup>           |           |           |           |           |           |           |         |
| Number positive                  | 14        | 7         | 9         | 17        | 10        | 11        | 68      |
| Mean EPG <sup>c</sup>            | 174       | 199       | 631       | 337       | 439       | 214       | 323     |
| POC-CCA <sup>b</sup>             |           |           |           |           |           |           |         |
| Number positive (trace positive) | 2         | 3         | 2         | 2         | 10        | 3         | 22      |
| Number positive (1+ or greater)  | 14        | 5         | 9         | 15        | 3         | 11        | 57      |
| Number negative                  | 4         | 4         | 5         | 5         | 8         | 9         | 35      |

#### Table 1. Demographic and epidemiologic characteristics of six neighborhoods in Azaguié Ahoua, south Côte d'Ivoire.

<sup>a</sup>Data from quadruplicate Kato-Katz thick smears

<sup>b</sup>Data from single POC-CCA urine cassette test

<sup>c</sup>Arithmetic mean EPG calculated from infected individuals.

doi:10.1371/journal.pntd.0004894.t001

correlated with EPG and was negatively associated with larger pool size, particularly at lower infection intensities.

#### Performance of the Urine Pooling Strategy and Traditional Stool Microscopy in Prevalence Classification Using the Microsimulation Model

We used a microsimulation to evaluate the diagnostic performance of three urine pooling strategies (n = 4, 8, and 12 pooled samples) to give a binary prediction for informing preventive chemotherapy programs in simulated cohorts across a prevalence range from 0 to 20%. The three pooling strategies gave comparable, and often superior, classification performance to traditional stool microscopy for the same number of tests (Figs <u>2</u> and <u>3</u>). As expected, classification error for all strategies was highest when the true prevalence was near the prevalence threshold, and classification improved further away from the prevalence threshold. For 80% and 90% certainty of correct classification of communities (± 5% around the 10% prevalence threshold), traditional stool microscopy required 71 and 150 tests, while urine pooling (n = 4) needed 33 and 67 tests, and urine pooling (n = 8) needed 29 and 73 tests, respectively.

#### Total Cost and Number of Tests with Urine Pooling Strategy

The urine pooling strategies (n = 4, 8, and 12) reduced the number of tests, while achieving the same accuracy as traditional stool microscopy across the full range of certainties in classification (Fig.3). Only the pooling strategy (n = 4) demonstrated cost savings compared to traditional

| Kato-Katz (quadruplicate) |                          | POC-CCA test             |  |  |  |  |
|---------------------------|--------------------------|--------------------------|--|--|--|--|
| POC-CCA(tr+)              |                          |                          |  |  |  |  |
| Sensitivity               | 78.2 (95% CI: 71.0-84.5) | 95.1 (95% CI: 89.2–98.5) |  |  |  |  |
| Specificity               | 96.1 (95% CI: 90.8–98.8) | 82.9 (95% Cl: 71.5–91.6) |  |  |  |  |
| POC-CCA(tr-)              |                          |                          |  |  |  |  |
| Sensitivity               | 78.9 (95% CI: 71.5–85.4) | 74.4 (95% Cl: 63.8–83.7) |  |  |  |  |
| Specificity               | 95.4 (95% CI: 89.2–98.6) | 94.5 (95% Cl: 86.9–98.5) |  |  |  |  |

Note: For microsimulation, Kato-Katz sensitivity was estimated from literature as a function of infection intensity (EPG)

doi:10.1371/journal.pntd.0004894.t002

| Test characteristic | Category                      | Number | Dilution, n = 4 | Dilution, n = 8 | Dilution, n = 12 |
|---------------------|-------------------------------|--------|-----------------|-----------------|------------------|
| Sensitivity         | Overall                       | 78     | 85.9            | 79.5            | 65.4             |
| POC-CCA(tr+)        | Light infection               | 43     | 76.7            | 69.8            | 48.8             |
|                     | Moderate infection            | 16     | 93.8            | 93.8            | 87.5             |
|                     | Heavy infection               | 19     | 100.0           | 89.5            | 84.2             |
| Sensitivity         | Overall                       | 78     | 61.5            | 47.4            | 30.8             |
| POC-CCA(tr-)        | Light infection               | 43     | 41.9            | 30.2            | 4.6              |
|                     | Moderate infection            | 16     | 81.3            | 68.8            | 62.5             |
|                     | Heavy infection               | 19     | 89.5            | 68.4            | 63.2             |
| Specificity         |                               |        |                 |                 |                  |
| POC-CCA(tr+)        | Tested <sup>a</sup>           | 10     | 90.0            | 100.0           | 100.0            |
|                     | Model conditions <sup>b</sup> | -      | 77.2            | 75.9            | 78.6             |
| Specificity         |                               |        |                 |                 |                  |
| POC-CCA(tr-)        | Tested <sup>a</sup>           | 10     | 100.0           | 100.0           | 100.0            |
|                     | Model conditions <sup>b</sup> | -      | 94.0            | 94.9            | 97.7             |

#### Table 3. Sensitivity and specificity of the POC-CCA test with a urine pooling strategy.

<sup>a</sup>Tested specificity was calculated using urine pools that were each negative by single POC-CCA test (see <u>S1 Text</u>).

<sup>b</sup>Model conditions specificity was calculated using the specificity of single POC-CCA with the assumption that the dilution effect upon false positives would mimic very light intensity infections (see <u>Methods</u> and <u>S1 Text</u>).

doi:10.1371/journal.pntd.0004894.t003

stool microscopy. The pooling strategies (n = 8 and 12) did not reduce the total cost. This result remained robust when evaluating the number of tests and total cost per correctly classified school (<u>S1 Text</u>).

#### **Results from Sensitivity Analysis**

The one-way sensitivity analyses found that our primary finding–urine pooling strategy (n = 4) yielded cost savings when compared to stool microscopy–was robust in the majority of alternative



**Fig 1.** Modeled estimation of the sensitivity of the urine pooling strategy. Positive urine sample confirmed by quadruplicate Kato-Katz thick smears and single POC-CCA urine cassette test were combined with three (n = 4), seven (n = 8), or 11 (n = 12) negative urine samples. A logistic regression was used to model sensitivity of the pooled urine samples (n = 4, 8, and 12) as a function of the infection intensity (EPG) of the one infected urine. The distribution of *S. mansoni* infections used in this model included light, moderate, and heavy intensity infections (figure inset).

doi:10.1371/journal.pntd.0004894.g001



**Fig 2.** Operating characteristic curves for microsimulation analysis of urine pooling strategy to rapidly map the prevalence of schistosomiasis and inform preventive chemotherapy. Using primary data on urine pooling strategies (n = 4, 8, and 12 samples), we used a microsimulation to model the proportion (Pr) of correct binary classification around a prevalence threshold to indicate need for school-based preventive chemotherapy according to WHO (10% prevalence). The interpretation of the curves is provided in (a). The results are presented for: (b) 20 tests; (c) 50 tests; and (d) 250 tests. We compared traditional stool microscopy (duplicate Kato-Katz thick smears derived from one stool sample), single POC-CCA test, and the three urine pooling strategies (pool sizes of 4, 8, and 12) using the WHO 10% prevalence threshold. The strategies performed well within the center of a prevalence categorization, but poorly at the boundary between two prevalence categories. Increased number of tests resulted in lower classification error.

doi:10.1371/journal.pntd.0004894.g002

assumption on pooling cost, setting-specific epidemiologic differences, sensitivity, and specificity except on the upper ranges ( $\underline{Fig 4}$ ). Under some assumptions, pooling 8 samples also yielded cost savings. The overall study findings were comparable when using a prevalence threshold of 5% ( $\underline{S1}$  Text).

#### Discussion

This study found that a urine pooling strategy using the commercialized POC-CCA test could be more efficient than individual-based surveys with traditional stool microscopy in informing where preventive chemotherapy against schistosomiasis is needed in low-endemicity settings. We characterized the sensitivity and specificity of urine pooling with the POC-CCA test across



**Fig 3.** Number of tests and total costs per school from microsimulation analysis of urine pooling strategy and traditional stool microscopy. The urine pooling strategy (n = 4, 8, and 12 pool) was compared against stool microscopy to estimate (a) the total number of tests per school and (b) total cost per school for identical level of certainty in binary classification on need for preventive chemotherapy. We used a 10% prevalence threshold in this base case analysis. A loess algorithm was applied for visualization.

doi:10.1371/journal.pntd.0004894.g003



**Fig 4. One-way sensitivity analysis of microsimulation.** This analysis tested the effect of changing individual model parameters on the total cost of the urine pooling strategy for (a) pool of 4 and (b) pool of 8. The horizontal bar represents the total cost to achieve 90% level of certainty (±5% around the prevalence threshold) in classification across a range of values for the tested parameter. The y-axis (solid black line) represents the total cost of the urine pooling strategy under base case assumptions. The vertical dashed blue line represents the total cost of traditional stool microscopy under base case assumptions, and all strategies to the left of this line indicate a cost saving advantage of the urine pooling strategy.

doi:10.1371/journal.pntd.0004894.g004

multiple dilutions, and used a statistical classification tool (lot quality assurance sampling) to operationalize this pooling strategy as a binary predictor of whether or not preventive chemotherapy is needed in low-endemicity settings. While sensitivity declined, as expected, at higher pool sizes, this loss was offset by the efficiency gains in screening multiple samples simultaneously. Through simulation modeling, we found that the pooling strategy reduced the number of tests and total cost, while achieving the same performance as traditional stool microscopy. These findings support the need for further validation of the urine pooling strategy in lowendemicity and near-elimination settings as a rapid, cost-saving alternative to traditional stool microscopy.

As the global strategy shifts from morbidity control to a goal of disease elimination and treatment programs are expanded, high-resolution mapping of where schistosomiasis is prevalent is crucial, especially since this disease is highly geographically focal [4, 23]. Additionally, some settings will reduce their prevalence to below the 10% threshold in the school-aged child population, and hence, preventive chemotherapy would no longer be indicated. Once settings do approach elimination, rigorous monitoring and surveillance will be essential to detect disease rebound. For all these reasons, a rapid and inexpensive approach to inform settings on the need for preventive chemotherapy will be crucial.

We evaluated the diagnostic performance and cost of a pooled urine methodology with multiple pool sizes (n = 4, 8, and 12). The use of sample pooling for diagnostic screening of parasitic and other infectious diseases has been previously demonstrated [8, 9]. We found that the urine pool size of 4 performed optimally with a balance of good sensitivity and specificity and relatively low cost. Across a range of certainties in classification, the urine pooling strategy (n = 4) reduced the number of tests and total cost with the same performance as traditional stool microscopy. The pool sizes of 8 and 12 had lower sensitivity, and ultimately offered limited test or cost savings except under select conditions. In this analysis, we treated POC-CCA trace positive test results as negative, which resulted in lower sensitivity but higher specificity. While treating trace positive results as positive greatly improved sensitivity, this resulted in substantially more false positives, which was compounded by the effect of sample pooling. We also evaluated single POC-CCA with the latent class model, which does not assume any diagnostic 'gold' standard test, and found our results for sensitivity and specificity in broad agreement with previous studies [5, 12, 19, 24]. Notably, our POC-CCA specificity estimate was lower than previous estimates making our modeling results conservative [5, 12, 24].

The LQAS classification tool–which is used for binary categorization based on a predefined threshold-is a useful method to reduce sampling effort and maintain accuracy [13, 14]. Our study focused on providing a binary classification around the 10% prevalence threshold (within  $\pm 5\%$  prevalence) for schistosomiasis, which is the threshold that schoolbased preventive chemotherapy is recommended. We did not assess multiple category-LQAS since the pooling strategy is best suited for binary classification, and the 10% threshold allowed for identification of settings where treatment would be indicated. Rigorous study has been given to the application of LQAS in classification of helminth prevalence [13, 14], although previous studies have assumed perfect sensitivity and specificity and estimated a sample size of 15 tests for 80% certainty ( $\pm$  6.5% around the 10% prevalence threshold). Under these conditions, our microsimulation corroborated the estimated sample size. However, when we incorporated imperfect diagnostics, we found that an increased sample size was needed. For 80% and 90% certainty, traditional stool microscopy required 71 and 150 tests, while urine pooling (n = 4) needed 33 and 67 tests. This suggests the importance of accounting for imperfect test characteristics and random sampling in LQAS calculations, and the utility of urine pooling to decrease the number of necessary tests. Notably, sampling effort for both traditional stool microscopy and urine pooling with the LQAS tool is still lower than

current WHO recommendations that suggest 250-500 school-aged children (50 per school) [2, 3, 14]. This demonstrates the value of LQAS to decrease sampling effort, total cost, and time necessary to correctly classify a setting.

We evaluated a sample pooling strategy with the POC-CCA urine cassette tests, which is designed to detect *S. mansoni* and performs poorly for detection of *S. haematobium* [25], although *S. haematobium* is often geographically overlapping and can co-infect individuals in sub-Saharan Africa [26]. Since a sample pooling strategy necessitates a test with high sensitivity, we focused our study in settings endemic with *S. mansoni*. Future work can examine incorporation of the circulating anodic antigen (CAA) test or polymerase chain reaction (PCR)-based methods that can detect multiple species of *Schistosoma* to provide a comprehensive strategy [27–31]. Molecular-based (e.g., PCR) techniques are an attractive option for use in pooling strategies because of their high sensitivity (>90%) and specificity, although current methods may be too costly and require advanced laboratory infrastructure that are often out of reach in resource-constrained settings where schistosomiasis is endemic [28–31].

The findings of this investigation should be understood within the limitations of the study design and model assumptions. We simulated theoretical settings by deriving a generalized epidemiologic relationship between prevalence, infection intensity, and parasite dispersion within the population based upon real data [21]. We assumed a negative binomial distribution of disease, which is based upon empiric observation and common practice [21, 22, 32, 33]. To calculate the specificity of the urine pooling strategy, we conservatively assumed that false positives would dilute identically to true positives, although future work should investigate this assumption. While the cost of pooling is uncertain, we assumed each extra pooled sample would incur an additional US\$ 0.50 cost and varied this in a sensitivity analysis with our primary findings remaining robust. We defined certainty as the proportion of correct classification  $\pm 5\%$  prevalence around the prevalence threshold following common practice, although this threshold for accuracy may be modified. A pooling strategy is best poised to be used in low prevalence settings since this will optimize reductions in number of tests and total cost, but in our study we obtained samples from moderate and high endemicity setting. This was to ensure a sufficient number of positive samples were collected, and to estimate the intrinsic relationship between sensitivity and EPG. We included samples from a wide range of infection intensities, especially light infections. We then modeled a broad range of prevalences (and associated EPG distributions) to capture the epidemiology of a low-endemicity setting. Finally, we did not address the potential for semi-curtailed or curtailed sampling, which is when sampling can be stopped early because of definite classification. However, while this offers an attractive option to reduce testing, often all samples are collected and tested simultaneously making this less practical, although future work could explore this option.

As treatment programs for control and elimination of schistosomiasis are expanded, new tools and strategies are needed to support the efficient targeting of preventive chemotherapy. Further field study of a urine pooling strategy employing POC-CCA cassette test for rapid and convenient *S. mansoni* diagnosis is warranted to validate this approach to support control and elimination of schistosomiasis.

#### **Supporting Information**

**S1 Text. Supplemental materials.** (DOCX)

**S1 Checklist. STARD checklist.** (DOCX)

#### **Author Contributions**

Conceived and designed the experiments: NCL JTC JK JRA IIB.

Performed the experiments: NCL JTC IIB.

Analyzed the data: NCL JRA.

Contributed reagents/materials/analysis tools: NCL JTC JK JRA IIB.

Wrote the paper: NCL JTC EKN EB JU JK IIB JRA.

Contributed intellectual content and approved final draft: NCL JTC EKN EB JU JK IIB JRA.

#### References

- Lai YS, Biedermann P, Ekpo UF, Garba A, Mathieu E, Midzi N, et al. (2015) Spatial distribution of schistosomiasis and treatment needs in sub-Saharan Africa: a systematic review and geostatistical analysis. Lancet Infect Dis. 15: 927–940. doi: 10.1016/S1473-3099(15)00066-3 PMID: 26004859.
- WHO (2006) Preventive chemotherapy in human helminthiasis: coordinated use of anthelminthic drugs in control interventions: a manual for health professionals and programme managers. Geneva: World Health Organization.
- 3. WHO (2011) Helminth control in school-age children: a guide for managers of control programmes. Geneva: World Health Organization.
- WHO (2012) Accelerating work to overcome the global impact of neglected tropical diseases: a roadmap for implementation. Geneva: World Health Organization.
- Lamberton PH, Kabatereine NB, Oguttu DW, Fenwick A, Webster JP (2014) Sensitivity and specificity of multiple Kato-Katz thick smears and a circulating cathodic antigen test for Schistosoma mansoni diagnosis pre- and post-repeated-praziquantel treatment. PLoS Negl Trop Dis. 8: e3139. doi: <u>10.1371/</u> journal.pntd.0003139 PMID: <u>25211217</u>.
- Worrell CM, Bartoces M, Karanja DM, Ochola EA, Matete DO, Mwinzi PN, et al. (2015) Cost analysis of tests for the detection of *Schistosoma mansoni* infection in children in western Kenya. Am J Trop Med Hyg. 92: 1233–1239. doi: 10.4269/ajtmh.14-0644 PMID: 25870422.
- Brooker S, Kabatereine NB, Gyapong JO, Stothard JR, Utzinger J (2009) Rapid mapping of schistosomiasis and other neglected tropical diseases in the context of integrated control programmes in Africa. Parasitology. 136: 1707–1718. doi: 10.1017/S0031182009005940 PMID: 19450373.
- Rinaldi L, Levecke B, Bosco A, Ianniello D, Pepe P, Charlier J, et al. (2014) Comparison of individual and pooled faecal samples in sheep for the assessment of gastrointestinal strongyle infection intensity and anthelmintic drug efficacy using McMaster and Mini-FLOTAC. Vet Parasitol. 205: 216–223. doi: 10.1016/j.vetpar.2014.06.011 PMID: 25002307.
- Kure A, Mekonnen Z, Dana D, Bajiro M, Ayana M, Vercruysse J, et al. (2015) Comparison of individual and pooled stool samples for the assessment of intensity of *Schistosoma mansoni* and soil-transmitted helminth infections using the Kato-Katz technique. Parasit Vectors. 8: 489. doi: <u>10.1186/s13071-015-</u> 1101-1 PMID: 26400064.
- Coulibaly JT, Knopp S, N'Guessan NA, Silué KD, Fürst T, Lohourignon LK, et al. (2011) Accuracy of urine circulating cathodic antigen (CCA) test for *Schistosoma mansoni* diagnosis in different settings of Côte d'Ivoire. PLoS Negl Trop Dis. 5: e1384. doi: <u>10.1371/journal.pntd.0001384</u> PMID: <u>22132246</u>.
- 11. van Dam GJ, Wichers JH, Ferreira TM, Ghati D, van Amerongen A, Deelder AM (2004) Diagnosis of schistosomiasis by reagent strip test for detection of circulating cathodic antigen. J Clin Microbiol. 42: 5458–5461. doi: <u>10.1128/JCM.42.12.5458-5461.2004</u> PMID: <u>15583265</u>.
- Coulibaly JT, N'Gbesso YK, Knopp S, N'Guessan NA, Silué KD, van Dam GJ, et al. (2013) Accuracy of urine circulating cathodic antigen test for the diagnosis of *Schistosoma mansoni* in preschool-aged children before and after treatment. PLoS Negl Trop Dis. 7: e2109. doi: <u>10.1371/journal.pntd.0002109</u> PMID: <u>23556011</u>.
- Brooker S, Kabatereine NB, Myatt M, Stothard JR, Fenwick A (2005) Rapid assessment of *Schisto-soma mansoni*: the validity, applicability and cost-effectiveness of the lot quality assurance sampling method in Uganda. Trop Med Int Health. 10: 647–658. doi: <u>10.1111/j.1365-3156.2005.01446.x</u> PMID: <u>15960703</u>.
- Olives C, Valadez JJ, Brooker SJ, Pagano M (2012) Multiple category-lot quality assurance sampling: a new classification system with application to schistosomiasis control. PLoS Negl Trop Dis. 6: e1806. doi: <u>10.1371/journal.pntd.0001806</u> PMID: <u>22970333</u>.

- Myatt M, Limburg H, Minassian D, Katyola D (2003) Field trial of applicability of lot quality assurance sampling survey method for rapid assessment of prevalence of active trachoma. Bull World Health Organ. 81: 877–885. PMID: <u>14997240</u>.
- Hajian-Tilaki K (2014) Sample size estimation in diagnostic test studies of biomedical informatics. J Biomed Inform. 48: 193–204. doi: <u>10.1016/j.jbi.2014.02.013</u> PMID: <u>24582925</u>.
- Katz N, Chaves A, Pellegrino J. (1972) A simple device for quantitative stool thick-smear technique in Schistosomiasis mansoni. Rev Inst Med Trop São Paulo. 14: 397–400. PMID: <u>4675644</u>.
- Joseph L, Gyorkos TW, Coupal L (1995) Bayesian estimation of disease prevalence and the parameters of diagnostic tests in the absence of a gold standard. Am J Epidemiol. 141: 263–272. PMID: 7840100.
- Shane HL, Verani JR, Abudho B, Montgomery SP, Blackstock AJ, Mwinzi PN, et al. (2011) Evaluation of urine CCA assays for detection of *Schistosoma mansoni* infection in western Kenya. PLoS Negl Trop Dis. 5: e951. doi: <u>10.1371/journal.pntd.0000951</u> PMID: <u>21283613</u>.
- Levecke B, Anderson RM, Berkvens D, Charlier J, Devleesschauwer B, Speybroeck N, et al. (2015) Mathematical inference on helminth egg counts in stool and its applications in mass drug administration programmes to control soil-transmitted helminthiasis in public health. Adv Parasitol. 87: 193–247. doi: 10.1016/bs.apar.2015.01.001 PMID: 25765196.
- Lo NC, Lai YS, Karagiannis-Voules DA, Bogoch II, Coulibaly JT, Bendavid E, et al. (2016) Assessment of global guidelines for preventive chemotherapy against schistosomiasis and soil-transmitted helminthiasis: a cost-effectiveness modelling study. Lancet Infect Dis. in press: doi: <u>10.1016/S1473-3099(16)</u> <u>30073-1</u> PMID: <u>27286968</u>.
- Lo NC, Bogoch II, Blackburn BG, Raso G, N'Goran EK, Coulibaly JT, et al. (2015) Comparison of community-wide, integrated mass drug administration strategies for schistosomiasis and soil-transmitted helminthiasis: a cost-effectiveness modelling study. Lancet Glob Health. 3: e629–638. doi: <u>10.1016/</u>S2214-109X(15)00047-9 PMID: 26385302.
- Colley DG, Bustinduy AL, Secor WE, King CH. (2014) Human schistosomiasis. Lancet 383: 2253– 2264. doi: 10.1016/S0140-6736(13)61949-2 PMID: 24698483.
- Silveira AM, Costa EG, Ray D, Suzuki BM, Hsieh MH, Fraga LA, et al. (2016) Evaluation of the CCA immuno-chromatographic test to diagnose *Schistosoma mansoni* in Minas Gerais state, Brazil. PLoS Negl Trop Dis. 10: e0004357. doi: <u>10.1371/journal.pntd.0004357</u> PMID: <u>26752073</u>.
- Ochodo EA, Gopalakrishna G, Spek B, Reitsma JB, van Lieshout L, Polman K, et al. (2015) Circulating antigen tests and urine reagent strips for diagnosis of active schistosomiasis in endemic areas. Cochrane Database Syst Rev (3: ): CD009579. doi: 10.1002/14651858.CD009579.pub2 PMID: 25758180.
- Chammartin F, Houngbedji CA, Hürlimann E, Yapi RB, Silué KD, Soro G, et al. (2014) Bayesian risk mapping and model-based estimation of *Schistosoma haematobium-Schistosoma mansoni* co-distribution in Côte d'Ivoire. PLoS Negl Trop Dis. 8: e3407. doi: <u>10.1371/journal.pntd.0003407</u> PMID: 25522007.
- Knopp S, Corstjens PLAM, Koukounari A, Cercamondi CI, Ame SM, Ali SM, et al. (2015) Sensitivity and specificity of a urine circulating anodic antigen test for the diagnosis of *Schistosoma haematobium* in low endemic settings. PLoS Negl Trop Dis. 9: e0003752. doi: <u>10.1371/journal.pntd.0003752</u> PMID: <u>25973845</u>.
- Cavalcanti MG, Silva LF, Peralta RH, Barreto MG, Peralta JM (2013) Schistosomiasis in areas of low endemicity: a new era in diagnosis. Trends Parasitol. 29: 75–82. doi: <u>10.1016/j.pt.2012.11.003</u> PMID: <u>23290589</u>.
- Enk MJ, Oliveira e Silva G, Rodrigues NB (2012) Diagnostic accuracy and applicability of a PCR system for the detection of *Schistosoma mansoni* DNA in human urine samples from an endemic area. PLoS One. 7: e38947. doi: 10.1371/journal.pone.0038947 PMID: 22701733.
- Sandoval N, Siles-Lucas M, Perez-Arellano JL, Carranza C, Puente S, Lopez-Aban J, et al. (2006) A new PCR-based approach for the specific amplification of DNA from different *Schistosoma* species applicable to human urine samples. Parasitology. 133: 581–587. doi: <u>10.1017/S0031182006000898</u> PMID: <u>16834820</u>.
- Vinkeles Melchers NV, van Dam GJ, Shaproski D, Kahama AI, Brienen EA, Vennervald BJ, et al. (2014) Diagnostic performance of *Schistosoma* real-time PCR in urine samples from Kenyan children infected with *Schistosoma haematobium*: day-to-day variation and follow-up after praziquantel treatment. PLoS Negl Trop Dis. 8: e2807. doi: 10.1371/journal.pntd.0002807 PMID: 24743389.
- 32. Anderson RM, May RM (1991) Infectious diseases of humans: dynamics and control. Oxford: Oxford University Press.
- Chan MS, Guyatt HL, Bundy DAP, Medley GF (1994) The development and validation of an age-structured model for the evaluation of disease control strategies for intestinal helminths. Parasitology. 109: 389–396. Epub 1994/09/01. PMID: <u>7970893</u>.